Author's response to reviews

Title: Perianal mantle cell lymphoma: a rare presentation mimicking external thrombosed hemorrhoid: a case report

Authors:

Baris Gulcu (baris_gulcu@hotmail.com)
Ali Ozer (draliozerr@gmail.com)
Hulya Ozturk Nazlioglu (hulyaozturk@uludag.edu.tr)
Ersin Ozturk (drozturk@uludag.edu.tr)
Tuncay Yilmazlar (tunyil@uludag.edu.tr)

Version: 5
Date: 27 November 2013

Author's response to reviews: see over
We revised and added the following information as pointed in your letter:

1. **Please include the Ethnicity of the patient in the Abstract and Case presentation sections:** He was turkish and it was written in both sections.

2. **Please include a Consent section in the manuscript stating that consent has been obtained.** The editorial office may request copies of the informed consent documentation at any time. It has been added as the following form:

   Consent

   Written informed consent was obtained from the patient described above for publication of this case report and all accompanying images. A copy of the written consent is available for review by the editor-in-chief of this journal.

3. **Please include a list of abbreviations used in the manuscript and their meanings after the Conclusions section.** We have written this part as the following:

   Abbreviation

   SCC, squamous cell carcinoma; MCL, Mantle cell lymphoma; NHL, non-Hodgkin lymphoma; GI, gastrointestinal; MLP, Multiple lymphomatous polyposis'

4. **Please include an Authors Contributions section at the end of the manuscript, before the reference list.** We added this section as the following:

   Authors' Contributions

   BG analyzed and interpreted the patient data regarding to characteristics of the disorder. BG and AO were the major contributor in writing the manuscript. EO was the primary surgeon and had the main role in literature researching. HON was responsible for pathological review. TY was the mentor in all process and determined the pattern of case presentation. All authors read and approved the final manuscript.

5. **Please include an acknowledgement section at the end of the manuscript before the reference list:** we added an acknowledgement section as the following:

   Acknowledgements

   We would like to thank the patient for her cooperation and consent for permitting us to use her medical data and clinical photographs for this case report. (A written informed consent was obtained).